Author's response to reviews

Title: The risk of adverse pregnancy outcomes in women who are overweight or obese

Authors:

Chaturica Athukorala (chaturica.athukorala@gmail.com)
Alice R Rumbold (alice.rumbold@menzies.edu.au)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)
Kristyn J Willson (kristyn.willson@adelaide.edu.au)

Version: 4 Date: 2 September 2010

Author's response to reviews:

Dear Editor,

The following changes have been made to the manuscript as per our recent correspondence.

Major revisions (we require the author to make these changes)
- the list of authors in the manuscript now conforms with the preferred style.
- its indicated in the 'Author’s contribution' segment that all four authors have read and approved the final manuscript.
- the track changes have been removed.

Best wishes
Chaturica Athukorala